Author’s response to reviews

Title: Glucocorticoid induced adrenal insufficiency is common in steroid treated glomerular diseases - proposed strategy for screening and management

Authors:

Alvin Karangizi (alvin.karangizi@gmail.com)
May Al-Shaghana (m_alshaghana@hotmail.co.uk)
Sarah Logan (Sarah.Logan@uhb.nhs.uk)
Sherwin Criseno (Sherwin.Criseno@uhb.nhs.uk)
Rachel Webster (Rachel.Webster@uhb.nhs.uk)
Kristien Boelaert (K.Boelaert@bham.ac.uk)
Peter Hewins (peter.Hewins@uhb.nhs.uk)
Lorraine Harper (l.harper@bham.ac.uk)

Version: 2 Date: 22 Apr 2019

Author’s response to reviews:

Alvin H K Karangizi
Institute of Clinical Sciences,
College of Medical and Dental Services
University of Birmingham
Birmingham B15 2TT
UK

Editor in Chief
BMC Nephrology

Dear Tillie Cryer

We acknowledge your latest correspondence regarding our manuscript titled “Glucocorticoid induced adrenal insufficiency is common in steroid treated glomerular diseases – proposed strategy for screening and management”. We would like to again express gratitude for the time taken by yourself and the reviewers in appraising our manuscript. We have addressed all the further requested revisions and provide point-by-point responses as follows:
1. Corresponding author
We have noted that the corresponding authors' email address, provided on the title page, differs to the one on file in the editorial manager system. Please ensure that the corresponding author and their details match in the manuscript file and in the editorial manager system.

The corresponding author, Prof Lorraine Harper’s email address on the editorial manager system has been updated to her preferred correspondence email, l.harper@bham.ac.uk and now matches the manuscript.

2. Title Page - provide email address of all authors
Please include the email addresses for all authors on the title page. The corresponding author should still be indicated. Please also ensure these email addresses match the email addresses provided in the editorial manager system.

All the authors’ emails have been included on the title page and we have ensured that they match the email addresses on the editorial manager system

3. Authors Contributions - clarify who is KD
We have noted that the initials 'KD' are included in the Authors' Contributions. These initials do not match any of the listed authors on the manuscript. Please clarify who these initials belong to.

This was an error and should read KB referring to Dr Kristien Boelaert. This has been updated on the manuscript.

4. Provide figure legends
Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

The figure title/legends have been moved to a new section ‘Figure Legends’ after the References as instructed.

5. Clean copy
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

A clean copy has been prepared and uploaded with the figures uploaded separately as instructed

We look forward to your response

Yours sincerely

Dr Alvin H K Karangizi, MBChB (Hons), MRCP
Academic Clinical Fellow in Nephrology
University of Birmingham